Author's response to reviews

Title: Effect of n-3 polyunsaturated fatty acid on gene expression of the critical enzymes involved in homocysteine metabolism

Authors:

Tao Huang (taohuang83@gmail.com)
Mark L Wahlqvist (profmlw@nhri.org.tw)
Duo Li (duoli@zju.edu.cn)

Version: 4 Date: 17 January 2012

Author's response to reviews: see over
Editor's report
Title: Effect of n-3 polyunsaturated fatty acid on gene expression of the critical enzymes involved in homocysteine metabolism

Editor's report:
Formatting changes to be made:
-----------------------------------------------
Major revisions (we require the author to make these changes)

Abstract: This should be identical in the manuscript and the submission system. Currently alterations to the manuscript have not been applied to the submission system abstract.
Response: Done!

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line numbering: Please remove any line numbering from within the manuscript.
Response: Done!

Tracking/Highlighting: Please remove any tracking or highlighting from the manuscript.
Response: Done!

Text colour: Please ensure all text is black within the manuscript.
Response: Done!

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: Done!